Author's response to reviews

Title: The PPARGC1A Gly482Ser polymorphism is associated with left ventricular diastolic dysfunction in men

Authors:

Erik Ingelsson (erik.ingelsson@pubcare.uu.se)
Louise Bennet (louise.bennet@med.lu.se)
Martin Ridderstråle (martin.ridderstrale@med.lu.se)
Marianne Söderström (marianne.m.soderstrom@vgregion.se)
Lennart Råstam (lennart.rastam@med.lu.se)
Ulf Lindblad (ulf.lindblad@med.lu.se)

Version: 4 Date: 12 October 2008

Author's response to reviews: see over
October 12, 2008

Melissa Norton, MD  
BMC Cardiovascular Disorders, Editor-in-Chief  
BioMed Central Ltd  
Middlesex House  
34-42 Cleveland Street  
London W1T 4LB, UK

Dear Dr. Norton,

We appreciate your preliminary acceptance regarding our manuscript ‘The PPARC1A Gly482Ser polymorphism is associated with left ventricular diastolic dysfunction in men’ (MS: 3679997892102204). There were no comments from reviewers to respond to at this time, but we carefully revised the title page according to the editor’s instructions:

1. We removed authors qualifications.
2. We included e-mail addresses for all authors.
3. We removed the “sources of support” section from the title page.

We finally went through the manuscript check list one more time before uploading the cover letter and revised manuscript.

We hope that the revised manuscript is now acceptable for publication in BMC Cardiovascular Disorders. Looking forward to hear from you soon again.

With best wishes,

Ulf Lindblad, MD, PhD  
Department of Public Health and Community Medicine / Primary Health Care  
Hälsovetarbacken 7  
PO Box 454  
SE-405 30 Göteborg  
SWEDEN  
E-mail: ulf.lindblad@allmed.gu.se  
Phone: +46-31-7866831